Author's response to reviews

Title: The impact of overweight and obesity on health-related quality of life in childhood - results from an intervention study

Authors:

Nora Wille (n.wille@uke.uni-hamburg.de)
Michael Erhart (m.erhart@uke.uni-hamburg.de)
Christiane Petersen (christiane.petersen@mobydickhamburg.de)
Ulrike Ravens-Sieberer (u.ravens-sieberer@uke.uni-hamburg.de)

Version: 4 Date: 22 December 2008

Author's response to reviews: see over
Dear Dr. Norton,

I was delighted that you accepted our paper for publication and conducted the formatting changes according to your recommendations. I hope that everything is correct now. If there should be another problem with the manuscript it would be great if we could solve it tomorrow (December 23rd) since I will be in the office for the last time tomorrow and I will not come back before the 21st of January. Otherwise, in case that further correspondence is necessary, it would be helpful if you could send the e-mails to my co-author Michael Erhart in the meantime (m.erhart@uke.uni-hamburg.de). However, I tried to make sure that everything is correct now.

Thank you very much for the perfect cooperation.

Sincerely,

Nora Wille

Formatting changes

(1) The vertical elements in Table 1 as well as merged cells that cross rows were deleted.
(2) Table 1 is referenced first now.
(3) “List of abbreviations” was changed to “Abbreviations”.
(4) “Figures” was changed to “Figure legends”.
(5) Regarding the Figure cropping I am afraid that I am not able to minimise white space more than I already did. I hope that it works this way or that you might be able to make this change for me as it was mentioned for the minor revisions?